Author's response to reviews

Title: Comparative fixation stability of different fixation types of an open wedge high tibial osteotomy: a finite element analysis

Authors:

Maxim Golovakha (emin.aghayev@memcenter.unibe.ch)
Weniamin Orljanski (emin.aghayev@memcenter.unibe.ch)
Karl-Peter Benedetto (emin.aghayev@memcenter.unibe.ch)
Sergey Panchenko (emin.aghayev@memcenter.unibe.ch)
Philippe Büchler (emin.aghayev@memcenter.unibe.ch)
Philipp Henle (emin.aghayev@memcenter.unibe.ch)
Emin Aghayev (emin.aghayev@memcenter.unibe.ch)

Version: 2 Date: 17 February 2014

Author's response to reviews:

Dear colleagues,

Thank you for the initial assessment comments. We have included now the list of email addresses for all co-authors on the title page, and added the list of abbreviations, competing interests statement and author’s contributions.

As this finite element analysis did not included any human subject, but was based on finite element modeling, no research approval from the local ethical committee and no informed patient consents were required.

Kind regards

Emin Aghayev